Author's response to reviews

Title: Organization specific predictors of job satisfaction: findings from a Canadian multi-site quality of work life cross-sectional survey

Authors:

Paul Krueger (kruegerp@mcmaster.ca)
Kevin Brazil (brazilk@mcmaster.ca)
Lynne Lohfeld (lohfeld@mcmaster.ca)
H. Gayle Edward (edwarg@mcmaster.ca)
David Lewis (dlewis@stjosham.on.ca)
Erin Tjam (eytjam@stmaryshosp.on.ca)

Version: 5 Date: 11 Mar 2002

Dear Editor:

On behalf of my co-authors, I would like to thank you for accepting this manuscript for publication in BMC Health Services Research. As requested, we have made changes to the title (removed all unnecessary capitals); inserted a section for Competing Interests (none declared) and bolded sections in the tables rather than include shading.

In addition, the following minor changes were made: 1) the St. Joseph's Health Care System (SJHCS) was recently renamed the St. Joseph's Health System (SJHS) - we have updated this name throughout the document, and 2) commas were added in Table 1 numbers (e.g. 2,941 rather than 2941).

Dr. Paul Krueger